Author's response to reviews

Title: Sex, age, deprivation and patterns in life expectancy in Quebec, Canada: a population-based study

Authors:

Nathalie Auger (nathalie.auger@inspq.qc.ca)
Carolyne Alix (carolyne.alix@inspq.qc.ca)
Geng Zang (geng.zang@umontreal.ca)
Mark Daniel (mark.daniel@unisa.edu.au)

Version: 2 Date: 16 March 2010

Author's response to reviews:

To the editor:

We are submitting the galley proofs for our manuscript. As requested in your email, we have reformatted Table 2 which now fits on an A4 Portrait page. We have removed the authors' qualifications and job titles from the manuscript, and have checked for any typographical errors.

We hope you are satisfied with the revisions, and are available to make any additional changes that are needed, if any.

Sincerely,
The authors